

# 

**Citation:** Senyonjo L, Addy J, Martin DL, Agyemang D, Yeboah-Manu D, Gwyn S, et al. (2021) Surveillance for peri-elimination trachoma recrudescence: Exploratory studies in Ghana. PLoS Negl Trop Dis 15(9): e0009744. https://doi.org/ 10.1371/journal.pntd.0009744

Editor: Richard Reithinger, RTI International, UNITED STATES

Received: May 19, 2021

Accepted: August 18, 2021

Published: September 20, 2021

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

**Data Availability Statement:** All relevant data are within the manuscript.

**Funding:** Sightsavers received financial support for this work from the Coalition for Operational Research on Neglected Tropical Diseases (grants NTD-SC 150D & NTD-SC 055), which is funded at The Task Force for Global Health by the Bill & Melinda Gates Foundation, the United States Agency for International Development through its Neglected Tropical Diseases Program; and UK aid from the British people. The funders had no role in **RESEARCH ARTICLE** 

# Surveillance for peri-elimination trachoma recrudescence: Exploratory studies in Ghana

Laura Senyonjo<sup>1,2\*</sup>, James Addy<sup>3</sup>, Diana L. Martin<sup>4</sup>, David Agyemang<sup>5</sup>, Dorothy Yeboah-Manu<sup>6</sup>, Sarah Gwyn<sup>4</sup>, Benjamin Marfo<sup>7</sup>, Adwoa Asante-Poku<sup>6</sup>, Agatha Aboe<sup>5</sup>, Ernest Mensah<sup>8</sup>, Anthony W. Solomon<sup>2</sup>, Robin L. Bailey<sup>2</sup>

1 Research Team, Sightsavers, Haywards Heath, United Kingdom, 2 Clinical Research Department, London School of Hygiene & Tropical Medicine, London, United Kingdom, 3 Eye Health Department, Ghana Health Service, Accra, Ghana, 4 Division of Parasitic Diseases and Malaria, Centers for Disease Control and Prevention, Atlanta, United States of America, 5 Sightsavers Ghana, Accra, Ghana, 6 Bacteriology Department, Noguchi Memorial Institute for Medical Research, Accra, Ghana, 7 Neglected Tropical Diseases Division, Ghana Health Service, Accra, Ghana, 8 NTD team, FHI360, Accra, Ghana

\* lsenyonjo@sightsavers.org

# Abstract

# Introduction

To date, eleven countries have been validated as having eliminated trachoma as a public health problem, including Ghana in 2018. Surveillance for recrudescence is needed both pre- and post-validation but evidence-based guidance on appropriate strategies is lacking. We explored two potential surveillance strategies in Ghana.

# Methodology/principal findings

Amongst randomly-selected communities enrolled in pre-validation on-going surveillance between 2011 and 2015, eight were identified as having had trachomatous-inflammation follicular (TF) prevalence  $\geq$ 5% in children aged 1–9 years between 2012 and 2014. These eight were re-visited in 2015 and 2016 and neighbouring communities were also added ("TF trigger" investigations). Resident children aged 1–9 years were then examined for trachoma and had a conjunctival swab to test for *Chlamydia trachomatis (Ct)* and a dried blood spot (DBS) taken to test for anti-Pgp3 antibodies. These investigations identified at least one community with evidence of probable recent *Ct* ocular transmission. However, the approach likely lacks sufficient spatio-temporal power to be reliable.

A post-validation surveillance strategy was also evaluated, this reviewed the ocular *Ct* infection and anti-Pgp3 seroprevalence data from the TF trigger investigations and from the pre-validation surveillance surveys in 2015 and 2016. Three communities identified as having ocular *Ct* infection >0% and anti-Pgp3 seroprevalence  $\geq$ 15.0% were identified, and along with three linked communities, were followed-up as part of the surveillance strategy. An additional three communities with a seroprevalence  $\geq$ 25.0% but no *Ct* infection were also followed up ("antibody and infection trigger" investigations). DBS were taken from all residents aged  $\geq$ 1 year and ocular swabs from all children aged 1–9 years. There was evidence of transmission in the group of communities visited in one district (Zabzugu-Tatale).

study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

There was no or little evidence of continued transmission in other districts, suggesting previous infection identified was transient or potentially not true ocular *Ct* infection.

#### Conclusions/significance

There is evidence of heterogeneity in *Ct* transmission dynamics in northern Ghana, even 10 years after wide-scale MDA has stopped. There is added value in monitoring *Ct* infection and anti-*Ct* antibodies, using these indicators to interrogate past or present surveillance strategies. This can result in a deeper understanding of transmission dynamics and inform new post-validation surveillance strategies. Opportunities should be explored for integrating PCR and serological-based markers into surveys conducted in trachoma elimination settings.

#### Author summary

The goal for trachoma programmes is elimination of trachoma as a public health problem. This means that ongoing low-level eye-to-eye transmission of the causative bacterium, *Chlamydia trachomatis* (*Ct*), is acceptable. Countries need to implement a suitable surveillance system to identify any return to higher transmission levels. The best methodology for doing this is not known. We first explored the approach used by Ghana in its standard programme, which involved monitoring a limited number of randomly selected communities for evidence of active (inflammatory) trachoma visible in children's eyes on examination by trained observers. Although this strategy led to identification of at least one community that had probably had recent *Ct* transmission, the approach is unlikely to consistently identify places where return to higher levels of transmission is a risk. We also explored using information on infection (detected in eye swabs) and antibodies to *Ct* (detected in the blood) to identify communities at risk. We found evidence of both persistent eye-to-eye *Ct* transmission and areas where infection was transient and has now gone away. We conclude that the use of infection and antibody data for surveillance of tra-choma appears promising.

# Introduction

Trachoma, caused by repeated conjunctival infection with the bacterium *Chlamydia trachomatis* (*Ct*), which can progress to trachomatous trichiasis (TT), the in-turning of the eyelids resulting in at least one eyelash touching the globe of the eye, is the leading infectious cause of blindness worldwide [1]. It is targeted for global elimination as a public health problem [2] using the "SAFE" strategy: Surgery to minimize progressive visual loss in individuals who have TT; mass drug administration [MDA] of <u>A</u>ntibiotics; and <u>F</u>acial cleanliness and <u>E</u>nvironmental improvement. Elimination as a public health problem can be validated if (1) there is evidence that specific disease prevalence thresholds have been achieved and maintained for a minimum of two years in the absence of antibiotic MDA in each formerly-endemic evaluation unit (EU) of a country and (2) a suitable system is in place to identify and manage incident TT cases [3]. The elimination prevalence thresholds are a trachomatous inflammation—follicular (TF) prevalence <5% in children aged 1–9 years and a TT prevalence <0.2% amongst  $\geq$ 15-year-olds [3].

Validation of elimination of trachoma as a public health problem does not require absence of ocular *Ct* infection [3–5]. Therefore, after antibiotic MDA has ceased, countries must monitor for potential recrudescence. However, there is no clear definition of trachoma recrudescence or re-emergence, including at what geographical level (community, sub-district or district) monitoring for it could or should be undertaken. While it may initially seem reasonable to define recrudescence as the transition of a district that had achieved a TF prevalence of <5% to subsequently having a TF prevalence  $\geq$ 5%, TF prevalence is a lagging indicator as *Ct* prevalence declines [6–9] and could also conceivably lag behind *Ct* prevalence if infection subsequently recrudesced. Waiting until TF prevalence is  $\geq$ 5% might therefore lead to missing opportunities for early intervention. Equally, the observation in some populations of moderate-to-high TF prevalence with low or absent ocular *Ct* infection raises a concern that other causes of follicular conjunctival inflammation might trigger an erroneous recrudescence alarm [10–14]. Either form of systematic misclassification (false negative or false positive) would be problematic for programmes.

The World Health Organization (WHO) guidance on methods for trachoma pre-validation surveillance has changed over time. In 2008, WHO recommended that surveillance for active trachoma be undertaken through examination of school-aged children for TF in 2–4 purposively selected sites per evaluation unit (EU). Sites in the "least developed and suspected most endemic" areas of the EU were to be selected [15]. In 2014, a WHO technical consultation [16] reviewed those recommendations and concluded that the methodology could be unduly influenced by bias and chance, compounded by an inadequate understanding of community-level factors that might be associated with *Ct* resurgence. The 2008 guidance was therefore replaced with a recommendation for pre-validation surveillance in each EU as a population-based surveillance survey carried out at least two years after the last impact survey [16]. There are no formal WHO guidelines regarding post-validation surveillance for active trachoma; this is a recommended activity of operational research, conducted within collaborations between health ministries and academic partners [3].

Designing operational research for post-validation surveillance is difficult because the evidence base for evaluation systems—when, where, who and what—is very limited. Several studies have examined use of serology and nucleic acid amplification tests (NAATs) as alternatives or adjuncts to TF for this purpose [4,17–22]. It may be possible to employ serology- and NAAT-based markers to identify communities at increased risk of recrudescence; this could begin pre- or post-validation [4], potentially in combination with surveillance activities for other diseases [23].

We present an evaluation of pre- and post-validation surveillance using serology and PCR, in the hope they can inform future refinement of trachoma programme surveillance strategies.

#### Methods

#### **Ethical considerations**

This research was approved by the Ghana Health Service (GHS) Ethical Review Committee (03/07/15 & 01/11/18) and the London School of Hygiene & Tropical Medicine Ethics Review Committee (reference: 10285 & 16169). Written informed consent was received from all participants and caregivers of minors, informed assent was also attained from children able to provide it. Individuals with TF were offered 1% tetracycline eye ointment. Individuals with TT were referred within the health system for management. This activity was reviewed by CDC and was conducted consistent with applicable federal law and CDC policy.

#### Study area

This evaluation was conducted in two previously trachoma-endemic regions of Ghana: Northern and Upper West. In 2008, all districts had achieved the TF elimination prevalence threshold and antibiotic MDA was stopped [24]. In 2018, WHO validated Ghana as having eliminated trachoma as a public health problem [25,26].

#### Study design

Two approaches for trachoma surveillance using clinical, antibody and infection indicators were evaluated to determine how effective they are in identifying on-going transmission and potential recrudescence.

- Evaluation of the on-going pre-validation surveillance system, implemented by GHS from 2011–2015. This strategy was based on assessment of the prevalence of TF in children aged 1–9 years in a limited number of randomly selected sites each year. We will refer to these investigations as the "TF trigger" component of our study; it was done prior to validation, in 2015–2016.
- 2. Identification of communities hypothesized to be at potential risk of recrudescence on the basis of serological and infection data from either the TF trigger investigations or a series of pre-validation surveillance surveys conducted by GHS in 2015–2016 [25]. These communities were followed up in 2019. This work explored a potential model for post-validation surveillance. We will refer to the 2019 investigations as the "infection and antibody trigger" component.

Fig 1 provides an overview of the study methodology and the various components.

#### TF trigger evaluation

From 2011–2015, GHS implemented an on-going pre-validation surveillance system based on then-current WHO recommendations [27] consisting of field visits to seven randomlyselected sites per district per year in 18 EUs: two communities and five primary schools or preschools and the communities in which attending children lived. At those sites, all participants aged 1–9 years were examined for TF and all available participating adults aged  $\geq$ 15 years were examined for TT. Once a school or community was visited as part of this process, it was removed from the sampling frame for all subsequent years. Through this methodology, from 2011–2015, a total of eight index communities were identified in which  $\geq$ 5% of examined children had TF. GHS subsequently examined children for trachoma in up to four communities neighbouring those index communities; no further communities were found in which  $\geq$ 5% of examined children had TF. Antibiotic MDA was undertaken in the eight index communities for three years [28]. The final MDA round was completed at least six months prior to the study investigations outlined below.

In 2015–2016, the eight index communities were each re-visited. For six of the eight communities, up to four neighbouring communities (identified by the health facility staff and based on proximity) were also visited. For the two districts (Sawla-Tuna-Kalba and Zabzugu-Tatale, in the northern region) in which more than one index community was identified, neighbouring communities were only chosen for the index community with the higher TF prevalence. In Sawla-Tuna-Kalba, the two index communities (Baale and Kema) are adjacent. All consenting individuals aged  $\geq 1$  year living in the index and neighbouring communities were examined for signs of trachoma and all children aged 1–9 years had a conjunctival swab and a dried blood spot (DBS) taken.



https://doi.org/10.1371/journal.pntd.0009744.g001

#### Infection and antibody trigger

In 2015–2016, GHS conducted a number of population-based pre-validation surveillance surveys [25]; DBS and ocular swabs were collected from children aged 1–9 years in six of the EUs, as outlined in Senyonjo *et al*, 2018 [4]. The infection and antibody trigger method, took a stepwise approach, reviewing available *Ct* infection and anti-Pgp3 antibody data from either the pre-validation surveillance surveys or from the TF trigger investigations. This identified three communities as having, in 1–9-year-olds, (1) a prevalence of ocular *Ct* infection >0% and (2) a prevalence of anti-Pgp3 antibodies  $\geq$ 15.0%. These three communities were included in the infection and antibody trigger investigations, with the expectation that they are the most at risk for potential recrudescence. For each of these communities, three communities "linked" to the index community were also invited to participate. The criteria considered for inclusion of a linked community were:

- Location within 5 km of the index community;
- Presence of a footpath or road directly linking it to the index community;
- Frequent interaction of residents with residents of the index community (based on report by residents of the index community), including (for example) through attendance of children at the same pre- or primary school.

Efforts were made to select linked communities from different compass points in relation to the index community.

Communities located in the same districts as the three index communities (Wa, Sawla-Tuna-Kalpa, Zabzugu-Tatale) that had a 2016 seroprevalence in 1–9-year-olds of  $\geq$ 25.0% but no *Ct* infection were also invited to participate.

In February 2019, all consenting individuals aged  $\geq 1$  year in selected communities were examined for signs of trachoma and had a DBS taken. Children aged 1–9 years also had a conjunctival swab taken. A maximum of 300 DBS were collected per community, so in larger communities, children aged 1–9 years were exhaustively sampled and individuals aged  $\geq 10$  years were only sampled in every nth household, with n being the sampling interval calculated to result in the 300 DBS being collected evenly throughout the sampling frame. Data were also collected by interview on socio-economic status, access to water and sanitation services, education levels, school attendance, number of persons living in the household and compound (where applicable), and social interactions between children that could facilitate transmission. Efforts were made to include individuals who were initially absent, with teams returning a minimum of once to any household where one or more residents were not present at the time of the team's first visit.

#### Data handling

Survey data were collected electronically using secure, Open Data Kit-based Android smartphone applications (LINKS, Task Force for Global Health, Atlanta GA, USA, <u>https://</u> <u>linkssystem.org</u>, for the TF trigger investigations; CommCare, Dimagi Inc., Cambridge MA, USA, for the infection and antibody trigger investigations). Data were uploaded to a cloudlocated server, which was password-protected with access only available to study investigators.

#### **Clinical assessment**

Participating subjects were examined for the five signs of trachoma in the WHO simplified system: TF, trachomatous inflammation—intense (TI), trachomatous scarring (TS), TT and

corneal opacity (CO)[29] by trained graders. Presence of pathology in one or both eyes was scored as that sign of trachoma.

All graders were ophthalmic nurses certified in 2015 using Global Trachoma Mapping Project methodologies [30,31], as previously described [25]. In 2019, graders were given refresher training on trachoma grading using photos; graders participated only if they achieved a kappa score of  $\geq$ 0.8 for TF in an inter-grader agreement test.

#### Sample collection

Swabs for *Ct* PCR were collected by passing a polyester-tipped swab three times across the upper tarsal conjunctiva of the left eyelid, using a standardised procedure described previously [4]. Precautions were taken to minimize field contamination of the swabs, including changing gloves after every sample taken. Negative control air swabs were taken by field teams after every 50 subjects for the TF trigger investigations and after every 25 subjects for the infection and antibody trigger investigations; the count restarted when the team moved to a new district. Samples were kept dry and cool in an ice box in the field.

To collect DBS, the participant's finger was cleansed using an alcohol-soaked swab before being pricked using a sterile single-use spring-loaded lancet. Blood was collected directly onto a filter paper wheel with six projections, each calibrated to absorb 10  $\mu$ l of blood. The filter paper wheel was air-dried in the shade before being individually packed in a sealable plastic bag and stored with desiccant.

Once collected, ocular swabs were stored at 4°C and transferred within a week on ice packs to Noguchi Memorial Institute for Medical Research (NMIMR), Accra, Ghana, where they were stored at -20°C until the time for analysis. The DBS were also stored at 4°C and (for the TF trigger investigations) transferred to NMIMR or (for the infection and antibody trigger investigations) sent to CDC, Atlanta, where they were stored at -20°C until being analysed.

#### Detection of ocular Ct infection and anti-Pgp3 antibody

PCR for both sets of samples was conducted at NMIMR. For the TF trigger investigations, detection of anti-Pgp3 antibodies was conducted at NMIMR using an enzyme-linked immunosorbent assay (ELISA) as previously described [4]. For the infection and antibody trigger investigations, antibodies were detected using the multiplex bead array (MBA) platform at CDC, described elsewhere [32,33].

Conjunctival swabs were eluted using sterile diethylpirocarbonate (DEPC) water and pooled into groups of five samples using a published pooling strategy [34]. They were then analysed for the presence of *Ct* DNA using the GeneXpert CT/NG Assay run on GeneXpert IV (Cepheid, Sunnyvale CA, USA). For positive pools, component samples were re-tested individually to identify specific positive(s). Indeterminate pools were re-tested using a new aliquot of each component sample and a new GeneXpert cartridge. Field control swabs were analysed individually. Two *Ct*-positive and two *Ct*-negative processing controls were run at the beginning of each week.

#### Statistical analysis

Statistical analyses were conducted using STATA (Version 17, StataCorp, College Station TX, USA) and R (R Core Team 2014). The threshold to determine a seropositive individual was calculated using a finite mixture model based on maximum likelihood methods, with a cut-off for positivity set as four standard deviations from the mean of the seronegative population [18]. For the MFI-bg values (data from the MBA) data were first log-transformed (log(x+1)), to accommodate zero values. For the samples analysed by ELISA, the cut-off was calculated as

1.091  $OD_{450nm}$ ; for the samples analysed by MBA, the cut-off for log (MFI-bg+1) was calculated as 5.724.

Fisher's exact test was used to assess the homogeneity of seroprevalence estimates between linked communities.

For the infection and antibody trigger investigations, simple associations between seropositivity in children aged 1–9 years and potential explanatory factors were evaluated using the Chi-square test or non-parametric test for trend, as applicable.

Seroconversion rates (SCR, the mean annual rate at which seronegative individuals become seropositive) were calculated using a simple reverse catalytic model (RCM) fitted to seroprevalence in one-year age groups (all ages  $\geq 1$  year), using maximum likelihood estimates [35]. Evidence for a change in SCR over time was conducted through comparisons of paired models, the first assuming constant transmission intensity, and the second assuming a change in transmission intensity at a specified time point [36].

For the infection and antibody trigger investigations, in communities in and around Mayido, spatial autocorrelation of household-level TF and seroprevalence (in children aged 1–9 years) within a village boundary was evaluated using Moran's I index.

#### Results

#### **TF trigger**

A total of 1,869 children aged 1–9 years were present at the time of the survey and enumerated from across 27 communities (8 index, 19 neighbouring, <u>Table 1</u>). All were examined for signs of trachoma and had DBS taken. 1,831 DBS were analysed and able to be matched to demographic data. Of individuals matched to DBS results, 49.0% were female. 1,773 children had an ocular swab taken.

Only one 1–9-year-old tested (in Kakalapere, Wa District) had a PCR-positive ocular swab. In this community, the *Ct* infection prevalence was 1.4%, the seroprevalence by ELISA was 16.2% and the TF prevalence was 5.4%, <u>Table 1</u>. The neighbouring community of Manliyiri had the highest seroprevalence recorded across the seven communities (28.6%). There was no evidence of a difference in the seroprevalence (p = 0.13) across sampled communities in Wa, Fig 2.

Data collected in 2015–2016. Base map: Made with Natural Earth. Free vector and raster map data @ naturalearthdata.com. https://www.naturalearthdata.com/downloads/10m-cultural-vectors/

There was an increase in seroprevalence by age, in Baale and neighbouring communities and likewise for Mampeasem and neighbouring communities, <u>Table 2</u>.

#### Infection and antibody trigger

The communities included were drawn from the TF trigger investigations described in the paper and the study published in Senyonjo et al, 2018 [37]. A total of 3,121 individuals were enumerated and present at the time of field teams' visits; 52.7% were female. A total of 3,050 DBS were taken from individuals aged  $\geq$ 1 year and 2,908 DBS were matched to demographic data. A total of 1,170 (37.5%) of those enumerated were aged 1–9 years, from whom 1,163 ocular swabs were taken and analysed.

One community (Mayido) had the highest prevalence of both Ct infection (1.6%) and anti-Pgp3 antibodies (14.4%) in 1–9-year-olds. Of the three cases of Ct infection identified in Mayido, two were from the same household. The community with the highest TF prevalence (Satinjado with 12.9% TF) was linked to Mayido, <u>Table 3</u>. Only one other community had any Ct infection—this community had only a single case of infection, no TF, and only 5.3% of

District	Community	TF prevalence 2011–2015*	Community type	n	TF prevalence in 2015– 2016 pre-validation survey (n positive)	<i>Ct</i> infection (n positive)	Anti-Pgp3 antibodies (ELISA platform) (n positive)	Chi-squared test (Fisher exact) for seroprevalence (X <sup>2</sup> ; p-value)
Sawla- Tuna- Kalba	Baale	6.8%	Index	56	0%	0%	16.1% (9)	15.8; 0.004
	Kema	6.7%	Index**	94	0%	0%	4.3% (4)	
	Soma		Neighbouring	140	0%	0%	2.1% (3)	
	Gbonbondouri		Neighbouring	119	0%	0%	9.2% (11)	
	Dani-uuri		Neighbouring	47	0%	0%	4.2% (2)	
West	Mampeasem	6.0%	Index	121	1.7% (2)	0%	2.5% (3)	6.1; 0.04
Gonja	Sakpege		Neighbouring	65	3.1% (2)	0%	0%	
	Lukula		Neighbouring	69	0%	0%	7.2% (5)	
Zabzugu-	Mangoase	7.8%	Index	58	1.7% (1)	0%	6.9% (4)	8.3; 0.03
Tatale	Jatoyili		Neighbouring	87	4.6% (4)	0%	1.2% (1)	
	Kpaligigbin		Neighbouring	44	0%	0%	13.6% (6)	
	Tasundo No 2		Neighbouring	62	0%	0%	6.5% (4)	
	Tamabug	6.1%	Index	60	1.7% (1)	0%	8.3% (5)	
Gushegu	Saguli	7.4%	Index	56	0%	0%	10.7% (6)	6.9; 0.11
	Galiwe		Neighbouring	60	3.3% (2)	0%	1.7% (1)	
	Tinyogu		Neighbouring	30	0%	0%	3.3% (1)	
	Zologu		Neighbouring	174	1.1% (2)	0%	3.5% (6)	
Wa	Kakalapere	6.6%	Index	74	5.4% (4)	1.4% (1)	16.2% (12)	7.0; 0.13
	Manliyiri		Neighbouring	14	0%	0%	28.6% (4)	
	Dariyiri		Neighbouring	43	2.3% (1)	0%	4.7% (2)	
	Bamkpama		Neighbouring	68	1.5% (1)	0%	11.8% (8)	
	Fooyiri		Neighbouring	14	0%	0%	7.1% (1)	
Nadowli	Moyiri	6.1%	Index	83	0%	0%	6.0% (5)	2.0; 0.68
	Nayikori		Neighbouring	89	3.4% (3)	0%	7.9% (7)	
	Wiche		Neighbouring	15	0%	0%	13.3% (2)	
	Dancheli		Neighbouring	55	0%	0%	7.3% (4)	
	Mwanwaare		Neighbouring	34	0%	0%	2.9% (1)	

Table 1. Clinical, serology and conjunctival *C. trachomatis* (*Ct*) infection data from children aged 1–9 years in index and neighbouring communities visited as part of TF trigger-based investigations.

\*As per the methodology used by GHS at the time, the figure represents the highest prevalence reported either in the left or right eye

\*\*Is in proximity to Baale and therefore is both an index and evaluated as a neighbouring community. Where there is more than one index community in a district, neighbouring communities are only selected for the index community with the highest TF prevalence.

TF: Trachomatous inflammation- follicular; Ct: Chlamydia trachomatis

https://doi.org/10.1371/journal.pntd.0009744.t001

children were positive for anti-Pgp3 antibodies, Table 3. There was no evidence of spatial autocorrelation of TF at household level in any of the communities in the district of Zabzugu-Tatale, where Mayido and Satinjado are located. Spatial autocorrelation of household seroprevalence was only identified in the community of Kogni (Moran's I -0.157; p = 0.05), also in Zabzugu-Tatale.

There was strong evidence (p<0.001) of a change in the SCR at a time point between 6 and 15 years prior to sample collection in each of the three groups of communities evaluated. For the groups of communities in Sawla-Tuna-Kalba and Wa Districts, the more recent SCR (SCR2) was significantly lower than the historic rate after the time point of the rate change (SCR1). In Mayido and surrounding communities of Zabzugu-Tatale District, the more recent SCR was higher, Table 4.





https://doi.org/10.1371/journal.pntd.0009744.g002

District Index (asterisks) and neighbouring communities		Non-parametric test for trend (z-score; p-value)		
Sawla-Tuna- Kalba	Baale*, Kema*, Soma, Gbonbondouri, Dani-uuri	2.35; 0.02		
West Gonja	Mampeasem*, Sakpege, Lukula	2.53; 0.01		
Zabzugu-Tatale	Mangoase*, Jatoyili, Kpaligigbin, Tasundo No 2	1.36; 0.18		
Gushegu	Saguli*, Galiwe,Tinyogu, Zologu	1.29; 0.20		
Wa	Kakalapere*, Manliyiri, Dariyiri, Bamkpama, Fooyiri	1.56; 0.12		
Nadowli	Moyiri*, Nayikori, Wiche, Dancheli, Mwanwaare	0.66; 0.51		

Table 2. Association between seropositivity and increasing age (1–9 years), TF trigger investigations.

https://doi.org/10.1371/journal.pntd.0009744.t002

District	Community	Community type*	n	TF prevalence (n positive)	Ct infection (n positive)	Anti-Pgp3 antibodies, MBA platform (n positive)	Chi-squared test (Fisher exact) for seroprevalence (X <sup>2</sup> ; p-value)	
Zabzugu- Tatale	Mayido (Kpaliba)	Index ( <i>Ct</i> infection)	194	4.6% (9)	1.6% (3)	14.4% (28)	14.5; 0.004	
	Satinjado	Linked	62	12.9% (8)	0%	3.2% (2)		
	Tilpado	Linked	127	6.3% (8)	0%	6.3% (8)		
	Kogni	Linked	149	4.7% (7)	0%	4.0% (6)		
Sawla-Tuna-	Kulwong no. 1	Index ( <i>Ct</i> infection)	38	0%	2.6% (1)	5.3% (2)	0.4; 1.0	
Kalba	Kulwong no. 2	Linked	38	7.9% (3)	0%	2.6% (1)		
	Chechereyili	Linked	36	0%	0%	2.8% (1)		
	Konhinyili	Linked	47	0%	0%	4.3% (2)		
	Kpali	Index (no infection)	79	2.5% (2)	0%	13.9% (11)	N/A	
Wa	Kakalapere	Index (Ct infection)	92	0%	0%	3.3% (3)	3.0; 0.84	
	Manliyiri	Index (no infection) & linked	17	0%	0%	0%		
	Deryiri	Linked	26	0%	0%	0%		
	Bamkpama	Linked	149	1.8% (8)	0%	3.4% (5)		
	Damwaateng	Linked	33	0%	0%	0%		
	Tuole	Index (no infection)	64	3.1% (2)	0%	6.3% (4)	N/A	

Table 3. Clinical, serology and ocular *C. trachomatis* (*Ct*) infection data from children aged 1–9 years in index and linked communities visited as part of the infection and antibody trigger investigations.

\**Ct* infection denotes that the community had *Ct* infection and an anti-Pgp3 seroprevalence of >15% in 2016.

No infection denotes the community had no Ct infection identified but an anti-Pgp3 seroprevalence of greater than 25% in 2016 (amongst children aged 1–9 years)

https://doi.org/10.1371/journal.pntd.0009744.t003

#### Discussion

This study explored two approaches to identify recrudescence of trachoma. It provides evidence of heterogeneity in a post-validation setting, with variation in transmission dynamics, as might have been expected, over a relatively small spatial scale.

Our TF trigger methodology identified one community (Kakalapere) with a 2015–2016 TF prevalence  $\geq$ 5%, anti-Pgp3 seroprevalence >15% and detectable ocular *Ct* infection and a neighbouring village with anti-pgp3 antibodies of >25%. When both communities were revisited in 2019, there was no evidence of persistent infection or recent *Ct* transmission. The evaluation of serological indicators as part of the pre-validation surveillance surveys conducted in Ghana in 2015–2016, which provided evidence that the country had achieved the elimination thresholds, reported an overall cluster-summarised mean anti-Pgp3 seroprevalence of 5.5% (95% CI: 4.8–6.3) by ELISA. In and around Baale, in Sawla-Tuna-Kalba, there was evidence of an increase in seropositivity with an increase in age. Considering the community of

Table 4.	Seroconversion rates	(SCR) within	communities sample	d, infection and antibo	dy trigger investigations
		<b>(</b>			

District	Index (and linked) communities	SCR 2 (most recent rate, after point of change)	SCR 1 (historical rate, before point of change)	Time point of change (years ago)	p-value
Zabzugu- Tatale	Mayido, (Satinjado, Tilpado, Kogni)	2.0 per 100 individuals per year (95%CI: 1.5–2.6)	0.4 per 100 individuals per year (95%CI: 0.3–0.7)	6	< 0.001
Sawla-Tuna- Kalba	Kulwong no1, (Kulwong no2, Chechereyili, Konhinyili)	0.9 per 100 individuals per year (95%CI: 0.4–1.5)	5.8 per 100 individuals per year (95%CI: 3.7–9.1)	11	< 0.001
Wa	Kakalapere, (Manliyiri, Deryiri, Bamkpama, Damwaateng)	0.6 per 100 individuals per year (95%CI: 0.4–0.9)	13.3 per 100 individuals per year (95%CI: 8.9–19.7)	15	< 0.001

https://doi.org/10.1371/journal.pntd.0009744.t004

Baale itself had a seroprevalence of over 15%, higher than the background seroprevalence rate, it could be argued that there is some evidence of past low-level transmission. However, this would be in line with the understanding of transmission dynamics in a post-validation setting and with no evidence of *Ct* infection, does not suggest recrudescence [37]. None of the other communities evaluated in the TF trigger work provided strong evidence of transmission within the previous nine years and none had current ocular *Ct* infection. While the TF trigger methodology was able to provide some evidence of limited *Ct* infection, it did not reveal significant onward transmission. A possible explanation for the lack of sustained transmission could be that there was transient infection in the community that was self-limiting, even if likely truncated as a result of the three additional rounds of MDA implemented prior to this study. Even so, the TF trigger approach likely lacks the spatio-temporal power to reliably identify communities experiencing recrudescence.

Evidence from the antibody and infection trigger data suggested that there has been persistent or repeated re-introduction of *Ct* infection over a three-year period in Mayido. While the *Ct* infection identified by serology could be ocular infection with urogenital *Ct* strains and therefore not indicative of trachoma, the convergence of indicators (infection, antibodies and disease) and household clustering of infections in two consecutive surveys suggests it was trachoma. The recent SCR increase seen in Mayido and its linked communities represents further evidence for on-going acquisition of anti-*Ct* antibodies and could be a warning sign of recrudescence. The trigger for programmatic action, whether that might be an absolute value or change in SCR alone, or one of those things in conjunction with exceeding a seroprevalence or infection threshold, still needs to be determined. It should be noted that observing an SCR increase here was only possible because we collected DBS from all ages, allowing for historical comparisons of transmission rates.

There are potentially other areas of northern Ghana with transmission dynamics similar to Mayido's but that were missed as a result of the random sampling methodology that we used. It is therefore important to better understand the geospatial risk factors associated with ongoing ocular *Ct* transmission in the post-elimination setting to guide programmes on where to focus resources for surveillance. We note that communities in and around Mayido differ from the surrounding population in having residents predominantly from the Konkomba ethnic group and mostly (66.7%) identifying as traditionalist. 2010 census data for Zabzugu district (split from Zabzugu-Tatale; containing the communities in question) [38] reported the main ethnic group of the district as Dagomba, with 36.0% of the population traditionalist. Further work will be required to understand spatial connectivity and the contextual factors that lead to re-introduction and facilitation of on-going *Ct* transmission between individuals [39]. Differences in social and behavioural practices, especially related to hygiene and sanitation, would be interesting to explore further.

It has been suggested that the distribution of infection in a peri-elimination setting would best fit an exponential curve, which has a heavy tail, and that therefore communities with higher infection prevalence would be occasionally expected. The implicit corollary is that community prevalence would regress to the mean over time [40]. Regression to the mean has been observed in hypo-endemic settings in Nepal, Tanzania and the Gambia, where infection disappeared in relatively-high-prevalence communities [8,41–44]. The present study demonstrates that this phenomenon can be either delayed (beyond 10 years) or even fail to occur. We identified an area in which Ct infection persisted, with transmission potentially increasing post-validation, many years after district-level MDA was stopped. Interruption of Ct transmission does not quickly and invariably follow attainment of the criteria for trachoma's elimination as a public health problem. If the goal of transmission interruption is to be proposed [45] it should be noted that it may not occur without additional measures to identify and address such foci. Some of the limitations of this study relate to the interpretation of test results. Infection and antibody tests do not distinguish ocular from urogenital *Ct* infection, and TF can arise from other local inflammatory stimuli [46,47]. The reverse catalytic model to calculate SCR may be an over-simplification of real-world transmission dynamics and estimates could be improved by using a model based on antibody levels and multiple time points [48]. There are also difficulties in evaluating community prevalence indicators in small communities, such as Manliyiri, which reported a large swing in seroprevalence from 28.6% in 2015 to 0% in 2019. Although every effort was made to reach all children aged 1–9 years, the omission of only a few children that may have been seropositive or the graduation (in age) of seropositive children over the time period, shifting them outside the sampling frame in 2019, would have a large impact on overall prevalence estimates.

Evidence of transmission between nearby communities could have important implications for wider recrudescence that may be of concern to programmes. The situation in and around Mayido is difficult to interpret, as although TF was identified across all four communities, there was varying anti-Pgp3 seroprevalence, ranging from 3.2% to 14.4% and generally low levels of *Ct* infection. There may have been low-prevalence *Ct* infection but it was not identified either because of the sensitivity of the test, the pooling process used for sample analysis or issues with swab collection or transport. It would be worthwhile to revisit this area after a minimum of two years, to evaluate if there is any evidence of an increase in ocular *Ct* transmission mandating programmatic intervention.

# Conclusion

There is evidence of heterogeneity in ocular Ct transmission dynamics after validation of trachoma elimination as a public health problem, even 10 years after wide-scale MDA has stopped. There is no suggestion from this study of a threat to elimination thresholds in Ghana. Elimination thresholds for trachoma are evaluated at the level of the district or administrative unit with a population of 100,000–250,000. This study has identified on-going transmission at a smaller geographic scale and the importance of and programmatic implications of such foci are still to be determined. However, such foci should be routinely observed to monitor potential recrudescence over a larger geographic scale. There is added value in monitoring anti-Ctantibodies and Ct infection, either within routine surveillance or to interrogate past strategies. This can result in a deeper understanding of transmission dynamics and inform new post-validation surveillance strategies. Opportunities should be explored for integrating infection and serology-based markers into other infectious disease surveillance surveys conducted in trachoma elimination settings.

# Acknowledgments

We are very grateful to Alexandre Chailloux (Sightsavers) for preparing Fig 1; our survey teams, field supervisors and laboratory assistants; and participating communities.

The authors alone are responsible for the views expressed in this article and they do not necessarily represent the views, decisions or policies of the institutions with which they are affiliated.

# **Author Contributions**

**Conceptualization:** Laura Senyonjo, Diana L. Martin, Agatha Aboe, Anthony W. Solomon, Robin L. Bailey.

Formal analysis: Laura Senyonjo, Diana L. Martin, Dorothy Yeboah-Manu, Adwoa Asante-Poku.

Funding acquisition: Laura Senyonjo.

- Investigation: Laura Senyonjo, James Addy, Ernest Mensah, Robin L. Bailey.
- Methodology: Laura Senyonjo, Anthony W. Solomon, Robin L. Bailey.
- Project administration: David Agyemang.
- Resources: James Addy, Diana L. Martin, Benjamin Marfo.
- **Supervision:** Laura Senyonjo, James Addy, David Agyemang, Dorothy Yeboah-Manu, Sarah Gwyn, Benjamin Marfo, Agatha Aboe, Ernest Mensah, Robin L. Bailey.
- Validation: Laura Senyonjo, Anthony W. Solomon, Robin L. Bailey.

Visualization: Laura Senyonjo.

- Writing original draft: Laura Senyonjo.
- Writing review & editing: James Addy, Diana L. Martin, David Agyemang, Dorothy Yeboah-Manu, Sarah Gwyn, Adwoa Asante-Poku, Anthony W. Solomon, Robin L. Bailey.

#### References

- Flaxman SR, Bourne RRA, Resnikoff S, Ackland P, Braithwaite T, Cicinelli MV, et al. Global causes of blindness and distance vision impairment 1990–2020: a systematic review and meta-analysis. Lancet Glob Health. 2017. https://doi.org/10.1016/S2214-109X(17)30393-5 PMID: 29032195
- World Health Assembly. Global elimination of blinding trachoma. 51st World Health Assembly, Geneva, 16 May 1998, Resolution WHA51.11. Geneva: World Health Organization; 1998.
- World Health Organization. Validation of elimination of trachoma as a public health problem (WHO/ HTM/NTD/2016.8). Geneva: World Health Organization; 2016.
- Senyonjo LG, Debrah O, Martin DL, Asante-Poku A, Migchelsen SJ, Gwyn S, et al. Serological and PCR-based markers of ocular *Chlamydia trachomatis* transmission in northern Ghana after elimination of trachoma as a public health problem. PLoS Negl Trop Dis. 2018; 12(12):e0007027. https://doi.org/ 10.1371/journal.pntd.0007027 PMID: 30550537
- Lietman TM, Pinsent A, Liu F, Deiner M, Hollingsworth TD, Porco TC. Models of Trachoma Transmission and Their Policy Implications: From Control to Elimination. Clin Infect Dis. 2018; 66(suppl\_4): S275–S80. https://doi.org/10.1093/cid/ciy004 PMID: 29860288
- Keenan JD, Lakew T, Alemayehu W, Melese M, Porco TC, Yi E, et al. Clinical activity and polymerase chain reaction evidence of chlamydial infection after repeated mass antibiotic treatments for trachoma. Am J Trop Med Hyg. 2010; 82(3):482–7. Epub 2010/03/09. <u>https://doi.org/10.4269/ajtmh.2010.09-0315</u> PMID: 20207878
- Lee JS, Munoz BE, Mkocha H, Gaydos CA, Quinn TC, West SK. The effect of multiple rounds of mass drug administration on the association between ocular *Chlamydia trachomatis* infection and follicular trachoma in preschool-aged children. PLoS Negl Trop Dis. 2014; 8(4):e2761. Epub 2014/04/12. https:// doi.org/10.1371/journal.pntd.0002761 PMID: 24722392
- Solomon AW, Harding-Esch E, Alexander ND, Aguirre A, Holland MJ, Bailey RL, et al. Two doses of azithromycin to eliminate trachoma in a Tanzanian community. N Engl J Med. 2008; 358(17):1870–1. https://doi.org/10.1056/NEJMc0706263 PMID: 18434662
- Solomon AW, Holland MJ, Alexander ND, Massae PA, Aguirre A, Natividad-Sancho A, et al. Mass treatment with single-dose azithromycin for trachoma. N Engl J Med. 2004; 351(19):1962–71. <u>https://doi.org/10.1056/NEJMoa040979</u> PMID: 15525721
- Burr SE, Hart JD, Edwards T, Baldeh I, Bojang E, Harding-Esch EM, et al. Association between ocular bacterial carriage and follicular trachoma following mass azithromycin distribution in The Gambia. PLoS Negl Trop Dis. 2013; 7(7):e2347. Epub 2013/08/13. <u>https://doi.org/10.1371/journal.pntd.0002347</u> PMID: 23936573
- 11. Butcher R, Sokana O, Jack K, Sui L, Russell C, Last A, et al. Clinical signs of trachoma are prevalent among Solomon Islanders who have no persistent markers of prior infection with *Chlamydia*

trachomatis. Wellcome Open Res. 2018; 3:14. https://doi.org/10.12688/wellcomeopenres.13423.2 PMID: 29588922

- Macleod CK, Butcher R, Mudaliar U, Natutusau K, Pavluck AL, Willis R, et al. Low Prevalence of Ocular *Chlamydia trachomatis* Infection and Active Trachoma in the Western Division of Fiji. PLoS Negl Trop Dis. 2016; 10(7):e0004798. Epub 2016/07/13. https://doi.org/10.1371/journal.pntd.0004798 PMID: 27404379
- Butcher RM, Sokana O, Jack K, Macleod CK, Marks M, Kalae E, et al. Low Prevalence of Conjunctival Infection with *Chlamydia trachomatis* in a Treatment-Naive Trachoma-Endemic Region of the Solomon Islands. PLoS Negl Trop Dis. 2016; 10(9):e0004863.
- Burr SE, Hart J, Samikwa L, Chaima D, Cooley G, Martin D, et al. Pgp3 seroprevalence and associations with active trachoma and ocular *Chlamydia trachomatis* infection in Malawi: cross-sectional surveys in six evaluation units. PLoS Negl Trop Dis. 2019; 13(10):e0007749. <u>https://doi.org/10.1371/journal.pntd.0007749</u> PMID: 31658258
- 15. World Health Organization. Report on the meeting on post-endemic surveillance for blinding trachoma. Geneva: World Health Organization; 2008.
- World Health Organization Strategic and Technical Advisory Group on Neglected Tropical Diseases. Technical consultation on trachoma surveillance. September 11–12, 2014, Task Force for Global Health, Decatur, USA (WHO/HTM/NTD/2015.02). Geneva: World Health Organization; 2015.
- 17. Pinsent A, Solomon AW, Bailey RL, Bid R, Cama A, Dean D, et al. The utility of serology for elimination surveillance of trachoma. Nature comms. 2018; 9(1):5444. Epub 2018/12/24.
- Migchelsen SJ, Sepulveda N, Martin DL, Cooley G, Gwyn S, Pickering H, et al. Serology reflects a decline in the prevalence of trachoma in two regions of The Gambia. Sci Rep. 2017; 7(1):15040. https:// doi.org/10.1038/s41598-017-15056-7 PMID: 29118442
- Last A, Burr S, Alexander N, Harding-Esch E, Roberts CH, Nabicassa M, et al. Spatial clustering of high load ocular *Chlamydia trachomatis* infection in trachoma: A cross-sectional population-based study. Pathog Dis. 2017. https://doi.org/10.1093/femspd/ftx050 PMID: 28472466
- Martin DL, Bid R, Sandi F, Goodhew EB, Massae PA, Lasway A, et al. Serology for trachoma surveillance after cessation of mass drug administration. PLoS Negl Trop Dis. 2015; 9(2):e0003555. Epub 2015/02/26. https://doi.org/10.1371/journal.pntd.0003555 PMID: 25714363
- Martin DL, Saboya-Diaz MI, Abashawl A, Alemayeh W, Gwyn S, Hooper PJ, et al. The use of serology for trachoma surveillance: Current status and priorities for future investigation. PLoS Negl Trop Dis. 2020; 14(9):e0008316. https://doi.org/10.1371/journal.pntd.0008316 PMID: 32970672
- World Health Organization Strategic and Technical Advisory Group for Neglected Tropical Diseases. Trachoma Alternative Indicators Study Data review, 31 August—1 September 2016, Geneva, Switzerland (WHO/HTM/NTD/PCT/2017.10). Geneva: World Health Organization; 2017.
- 23. Handley BL, Butcher R, Taoaba R, Roberts CH, Cama A, Muller A, et al. Absence of Serological Evidence of Exposure to *Treponema pallidum* among Children Suggests Yaws Is No Longer Endemic in Kiribati. Am J Trop Med Hyg. 2019. https://doi.org/10.4269/ajtmh.18-0799 PMID: 30719964
- Yayemain D, King JD, Debrah O, Emerson PM, Aboe A, Ahorsu F, et al. Achieving trachoma control in Ghana after implementing the SAFE strategy. Trans Royal Soc Trop Med Hyg. 2009; 103(10):993– 1000. Epub 2009/03/17. https://doi.org/10.1016/j.trstmh.2009.02.007 PMID: 19286239
- Debrah O, Mensah EO, Senyonjo L, de Souza DK, Hervie TE, Agyemang D, et al. Elimination of trachoma as a public health problem in Ghana: Providing evidence through a pre-validation survey. PLoS Negl Trop Dis. 2017; 11(12):e0006099. Epub 2017/12/13. <u>https://doi.org/10.1371/journal.pntd.0006099</u> PMID: 29232708
- 26. World Health Organization. Ghana eliminates trachoma, freeing millions from suffering and blindness [Press release, 13 June 2018]. Brazzaville, Accra & Geneva: World Health Organization; 2018.
- Senyonjo L, Aboe A, Bailey R, Agyemang D, Marfo B, Wanye S, et al. Operational adaptations of the trachoma pre-validation surveillance strategy employed in Ghana: a qualitative assessment of successes and challenges. Infect Dis Poverty. 2019; 8(1):78. <u>https://doi.org/10.1186/s40249-019-0585-x</u> PMID: 31455431
- Ministry of Health. Dossier documenting elimination of trachoma as a public health problem: Ghana (available at: http://espen.afro.who.int/system/files/content/resources/GhanaTrachomaDossier\_EN. pdf, accessed 09 July 2018). Accra: Government of Ghana; 2018.
- Thylefors B, Dawson CR, Jones BR, West SK, Taylor HR. A simple system for the assessment of trachoma and its complications. Bull World Health Organ. 1987; 65(4):477–83. PMID: 3500800
- Solomon AW, Pavluck A, Courtright P, Aboe A, Adamu L, Alemayehu W, et al. The Global Trachoma Mapping Project: methodology of a 34-country population-based study. Ophthalmic Epidemiol. 2015; 22(3):214–25. https://doi.org/10.3109/09286586.2015.1037401 PMID: 26158580

- Solomon AW, Willis R, Pavluck AL, Alemayehu W, Bakhtiari A, Bovill S, et al. Quality Assurance and Quality Control in the Global Trachoma Mapping Project. Am J Trop Med Hyg. 2018; 99(4):858–63. https://doi.org/10.4269/ajtmh.18-0082 PMID: 30039782
- 32. Goodhew EB, Priest JW, Moss DM, Zhong G, Munoz B, Mkocha H, et al. CT694 and Pgp3 as serological tools for monitoring trachoma programs. PLoS Negl Trop Dis. 2012; 6(11):e1873. Epub 2012/11/08. https://doi.org/10.1371/journal.pntd.0001873 PMID: 23133684
- Moss DM, Priest JW, Boyd A, Weinkopff T, Kucerova Z, Beach MJ, et al. Multiplex bead assay for serum samples from children in Haiti enrolled in a drug study for the treatment of lymphatic filariasis. Am J Trop Med Hyg. 2011; 85(2):229–37. https://doi.org/10.4269/ajtmh.2011.11-0029 PMID: 21813840
- Dize L, West SK, Mkocha H, Quinn TC, Gaydos CA. Evaluation of pooled ocular and vaginal swabs by the Cepheid GeneXpert CT/NG assay for the detection of *Chlamydia trachomatis* and *Neisseria gonorrhoeae* compared to the GenProbe Aptima Combo 2 Assay. Diagn Microbiol Infect Dis. 2015; 81 (2):102–4. https://doi.org/10.1016/j.diagmicrobio.2014.11.010 PMID: 25497459
- **35.** Sepulveda N, Stresman G, White MT, Drakeley CJ. Current Mathematical Models for Analyzing Anti-Malarial Antibody Data with an Eye to Malaria Elimination and Eradication. J Immun Res. 2015:738030. Epub 2016/01/16. https://doi.org/10.1155/2015/738030 PMID: 26770994
- Drakeley CJ, Corran PH, Coleman PG, Tongren JE, McDonald SL, Carneiro I, et al. Estimating medium- and long-term trends in malaria transmission by using serological markers of malaria exposure. Proc Nat Acad Sci USA. 2005; 102(14):5108–13. Epub 2005/03/29. <u>https://doi.org/10.1073/pnas.</u> 0408725102 PMID: 15792998
- Senyonjo LG, Debrah O, Martin DL, Asante-Poku A, Migchelsen SJ, Gwyn S, et al. Serological and PCR-based markers of ocular Chlamydia trachomatis transmission in northern Ghana after elimination of trachoma as a public health problem. PLoS Negl Trop Dis. 2018; 12(12):e0007027–e. <u>https://doi.org/ 10.1371/journal.pntd.0007027</u> PMID: 30550537
- Ghana Statistical Service. 2010 Population & Housing Census; district analytical report, Zabzugu district. Accra: GSS;2014.
- Lloyd-Smith JO. Vacated niches, competitive release and the community ecology of pathogen eradication. Philos Trans R Soc Lond B Biol Sci. 2013; 368(1623):20120150. Epub 2013/06/27. https://doi.org/ 10.1098/rstb.2012.0150 PMID: 23798698
- Lietman TM, Gebre T, Abdou A, Alemayehu W, Emerson P, Blumberg S, et al. The distribution of the prevalence of ocular chlamydial infection in communities where trachoma is disappearing. Epidemics. 2015; 11:85–91. Epub 2015/05/17. https://doi.org/10.1016/j.epidem.2015.03.003 PMID: 25979286
- Gaynor BD, Miao Y, Cevallos V, Jha H, Chaudary JS, Bhatta R, et al. Eliminating trachoma in areas with limited disease. Emerg Infect Dis. 2003; 9(5):596–8. https://doi.org/10.3201/eid0905.020577 PMID: 12737745
- Burton MJ, Holland MJ, Makalo P, Aryee EA, Sillah A, Cohuet S, et al. Profound and sustained reduction in *Chlamydia trachomatis* in The Gambia: a five-year longitudinal study of trachoma endemic communities. PLoS Negl Trop Dis. 2010; 4(10). Epub 2010/10/20. https://doi.org/10.1371/journal.pntd. 0000835 PMID: 20957147
- Jha H, Chaudary JS, Bhatta R, Miao Y, Osaki-Holm S, Gaynor B, et al. Disappearance of trachoma from Western Nepal. Clin Infect Dis. 2002; 35(6):765–8. https://doi.org/10.1086/342298 PMID: 12203176
- 44. Harding-Esch EM, Edwards T, Sillah A, Sarr I, Roberts CH, Snell P, et al. Active trachoma and ocular Chlamydia trachomatis infection in two Gambian regions: on course for elimination by 2020? PLoS Negl Trop Dis. 2009; 3(12):e573. Epub 2009/12/23. https://doi.org/10.1371/journal.pntd.0000573 PMID: 20027217
- Oldenburg CE, Aragie S, Amza A, Solomon AW, Brogdon J, Arnold BF, et al. Can we eradicate trachoma? A survey of stakeholders. BJO. 2020:bjophthalmol-2020-315815. <u>https://doi.org/10.1136/ bjophthalmol-2020-315815 PMID: 32855161</u>
- Burton MJ, Hu VH, Massae P, Burr SE, Chevallier C, Afwamba IA, et al. What is causing active trachoma? The role of nonchlamydial bacterial pathogens in a low prevalence setting. IOVS. 2011; 52 (8):6012–7. Epub 2011/06/23.
- Baral K, Osaki S, Shreshta B, Panta CR, Boulter A, Pang F, et al. Reliability of clinical diagnosis in identifying infectious trachoma in a low-prevalence area of Nepal. Bull WHO. 1999; 77(6):461–6. Epub 1999/07/31. PMID: 10427930
- Yman V, White MT, Rono J, Arca B, Osier FH, Troye-Blomberg M, et al. Antibody acquisition models: A new tool for serological surveillance of malaria transmission intensity. Sci Rep. 2016; 6:19472. https:// doi.org/10.1038/srep19472 PMID: 26846726